Author's response to reviews

Title: Liver cirrhosis, other liver diseases, and risk of hospitalisation for intracerebral haemorrhage: A Danish population-based case-control study

Authors:

Henning Gronbaek (henning.gronbaek@dadlnet.dk)
Søren P Johnsen (SPJ@dce.au.dk)
Peter Jepsen (PJ@dce.au.dk)
Mette Gislum (MG@dce.au.dk)
Hendrik Vilstrup (hvils@as.aaa.dk)
Ulrik Tage-Jensen (utj@rn.dk)
Henrik T Sørensen (HTS@dce.au.dk)

Version: 2 Date: 6 April 2008

Author's response to reviews:

Dear Hans Zauner, PhD, Senior Assistant Editor

Thanks for the swift handling of the MS.

In the revised version we have included the statements regarding ethics (material & methods section), 'Authors' Contributions' and 'Competing Interests' (after conclusions).

Yours Henning Grønbæk, M.D., Ph.D.